Author's response to reviews

Title: Randomised controlled trial examining the effect of exercise in people with rheumatoid arthritis taking anti-TNFalpha therapy medication

Authors:

   Angela Reid (angelatreid@hotmail.com)
   Audrey Brady (audbrady@yahoo.co.uk)
   Catherine Blake (c.blake@ucd.ie)
   Anne-Barbara Mongey (anne.b.mongey@ucd.ie)
   Douglas J Veale (douglas.veale@ucd.ie)
   Oliver FitzGerald (oliver.fitzgerald@ucd.ie)
   Tara Cusack (t.cusack@ucd.ie)

Version: 6 Date: 13 December 2010

Author's response to reviews: see over
<table>
<thead>
<tr>
<th><strong>Reviewers Recommendation</strong></th>
<th><strong>Authors Response</strong></th>
</tr>
</thead>
<tbody>
<tr>
<td>Hypotheses were still not two sided</td>
<td>The authors have phrased the hypotheses as two sided, as directed by the reviewer. The change has been inserted in red font (lines 208 and 215).</td>
</tr>
<tr>
<td>The second hypothesis should be called alternative hypothesis</td>
<td>This had been changed, as recommended, and highlighted in red font (line 213)</td>
</tr>
<tr>
<td>Statistics has to be clarified for the primary endpoint which should be named, here still the time point (8 or 26 weeks) has to be given</td>
<td>This has been clarified in line 446 (time points are both 8 and 24 weeks) and highlighted in red font.</td>
</tr>
</tbody>
</table>